Author’s response to reviews

Title: Influence of electromagnetic radiation emitted by daily-use electronic devices on the Eyemate® system in-vitro: a feasibility study

Authors:

Azzurra Invernizzi (a.invernizzi@umcg.nl)
Shereif Haykal (s.a.m.m.haykal@umcg.nl)
Valeria Lo Faro (v.lo.faro@umcg.nl)
Vincenzo Pennisi (vincenzo.pennisi@med.ovgu.de)
Lars Choritz (lars.choritz@med.ovgu.de)

Version: 2 Date: 18 Aug 2020

Author’s response to reviews:

- Assistant Editor Comments: Please review your additional files as they do not currently match our guidelines.

- Response: Our additional files were in MAT format, now there are in XLS format, which matches the journal's guidelines.